Author's response to reviews

Title: Migraine aura or TIA? A 5-year follow-up case-control study of women with transient CNS disorders in pregnancy.

Authors:

 Janne Marit JM Ertresvag (Janne.Ertresvag@helse-sunnmore.no)
 Lars Jacob LJ Stovner (lars.stovner@ntnu.no)
 Lene L Ekern Kvavik (leneekernkvavik@hotmail.com)
 Hans-Jorgen H-J Johnsen (Hans.Jorgen.Johnsen@stolav.no)
 John-Anker J-A Zwart (john-anker.zwart@ntnu.no)
 Grethe G Helde (grethe.helde@ntnu.no)
 Gunnar G Bovim (gunnar.bovim@ntnu.no)

Version: 3 Date: 8 June 2007

Author's response to reviews:

In accordance with requirements from the editor, we have now made the following changes:

1) In the Background section of the abstract we have added the following: "Aims of the present study were to investigate and diagnose all pregnant women with transient neurological disorders of suspected central nervous system origin, and to compare this group with a control group of pregnant women with regard to vascular risk factors and prognosis."

2) In the last paragraph of the "Materials and Methods section we state that "A signed (written) consent was given from all participants."